Author’s response to reviews

Title: Does attending an exercise class with a spouse improve long-term exercise adherence among people aged 65 years and older: A 6-month prospective follow-up study

Authors:

Yosuke Osuka (osuka@tmig.or.jp)
Songee Jung (jonsoni@ncgg.go.jp)
Taeho Kim (kth3442@gmail.com)
Yoshiro Okubo (y.okubo@neura.edu.au)
Eunbi Kim (eunbi1229jp@yahoo.co.jp)
Kiyoji Tanaka (tanaka@taiiku.tsukuba.ac.jp)

Version: 1 Date: 03 Nov 2016

Author’s response to reviews:

We are grateful to editor for a consideration that have helped us to improve our paper clearly. In accordance with the editor's comment, we retrospectively registered our trial.